Author's response to reviews

Title: Is the core-periphery labour market structure related to perceived health? Findings of the Northern Swedish Cohort

Authors:

Anna-Karin Waenerlund (annakarin.waenerlund@fammed.umu.se)
Per E Gustafsson (per.e.gustafsson@fammed.umu.se)
Pekka Virtanen (pekka.j.virtanen@uta.fi)
Anne Hammarström (anne.hammarstrom@fammed.umu.se)

Version: 8 Date: 21 December 2011

Author's response to reviews: see over
Dear Professor Cullen,

We thank you for accepting the manuscript *Is the core-periphery labour market structure related to perceived health? Findings of the Northern Swedish Cohort* (Ms. No. 1224737272581105).

Regarding the questions on the manuscript
1. The city ?Tampere? has been inserted for affiliation ?2?. Please verify this is appropriate or correct if necessary.
Response: Yes, the city is Tampere this is correct.
2. Citations and corresponding references in the list were renumbered accordingly. Please verify if this is appropriate.
Response: The reference list was not renumbered accordingly. In this new version the references are altered.
3. Please provide volume and page number of reference ?18?.
Response: Reference 18 is a E-publication ahead of print, and do not therefore yet have a volume and page number.

As a lot of changes had to be made in the manuscript [especially errors regarding the references] I kindly request to see the new version before publication to make sure everything is correct.
Kind regards,

Anna-Karin Waenerlund, Corresponding Author

Per E Gustafsson

Pekka Virtanen

Anne Hammarström